Title: Neurofibroma invading into urinary bladder presenting with symptoms of obstructed defecation and a large perineal hernia

Authors:

Duminda Subasinghe (dumindas1982@yahoo.com)
Chathuranga Keppetiyagama (Chathurangakap@gmail.com)
Chandu de Silva (desilva.chandu@yahoo.com)
Neville Perera (nevilledperera@yahoo.com)
Dharmabandhu N Samarasekera (samarasekera58@yahoo.co.uk)

Version: 10 Date: 3 March 2014

Author's response to reviews:

Revisions-

1. As requested, the title page was removed from the additional files to the beginning of the manuscript.

2. As requested, response to reviewers was removed from the additional files.

3. As requested, the legends were removed from uploaded files (i.e. pictures), and instead, included them as a list at the end of our manuscript document (i.e. below acknowledgements).